

Federal Reserve Bank of Chicago

Not Working: Demographic Changes, Policy Changes, and the Distribution of Weeks (Not) Worked

Lisa Barrow and Kristin F. Butcher

WP 2004-23

Comments welcome

Not Working: Demographic Changes, Policy Changes, and the Distribution of Weeks (Not) Worked

Lisa Barrow Federal Reserve Bank of Chicago

Kristin F. Butcher Federal Reserve Bank of Chicago

November 30, 2004

Note: We would like to thank participants at the "Micro lunch" workshop in the research department at the Chicago Fed, participants in the Federal Reserve's Micro Systems conference, Orley Ashenfelter, and participants at the SOLE meetings for helpful comments. We thank Chinhui Juhn for helping with the Current Population Survey sample. Thanks also to Katharine Anderson, Joan Coogan, and Kyung Park for exemplary research assistance. The views expressed here are those of the authors and do not necessarily represent those of the Federal Reserve Bank of Chicago or the Federal Reserve System.

Abstract

From 1978 to 2000 the fraction of adult men in full-year non-employment increased from 17.1 to 21.6 percent. Previous research focused on the role of disability insurance policy and wage structure changes to explain this increase. Using Current Population Surveys from 1979 to 2003 we assess how much of the changes in full-year non-employment can be explained by demographic changes, possibly linked to health. With our empirical strategy we examine how 1978 to 2000 changes in demographic characteristics would have changed the distribution of weeks worked if policies and macroeconomic conditions remained as they were in 1978. For prime-aged men, we find changes in age, race, and ethnicity can "explain" 14 to 33 percent of the increase in full-year non-employment, without any change in policy or wage structure. For prime-aged women, changes in demographics also would have predicted increases in full-year non-employment, when in fact we saw dramatic decreases.

I. Introduction

The decrease in men's labor market participation over the last two decades has garnered much attention among researchers and policy makers (see for example, Autor and Duggan (2003) and Juhn, Murphy and Topel (2002)). The reason for this attention is several-fold. From a microeconomic perspective, researchers are interested in understanding how individuals respond to changes in policy. From a macroeconomic perspective, significant changes in the fraction of the population that participates in the labor force make it more difficult to compare unemployment rates across time periods and thus infer information about the strength of the economy.

Recent work has focussed on changes in disability insurance eligibility criteria, other public policies, and wages for low-skilled workers as potential explanations for increases in men's nonparticipation in the labor force. Surprisingly little attention has been paid to demographic change as a potential driver of the increase in nonparticipation.¹ In particular, we know that baby boomers have grown older and that there have been increases in the share of the population that is African American or Hispanic. It is possible that these demographic changes in policy or labor market returns because these populations have, on average, worse health outcomes than young and/or white populations. Additionally, most of the focus in this literature has been on men, because it is their labor market participation that has been declining. Women's labor force participation has been increasing over the same period, raising important questions about whether demographic changes and changes in policies governing disability eligibility may have affected them differently.

In this paper we use the March Current Population Surveys from 1979 to 2003 to examine changes in the distribution of weeks worked per year for both men and women. We adapt techniques from DiNardo, Fortin, and Lemieux (1996) to examine questions such as: how would the demographic changes that occurred between 1978 and 2000 have changed the distribution of weeks worked if policies had remained as they were in 1978?² Here we focus on reported weeks worked, and in particular on individuals who report not working for the full year. We use actual work behavior as the outcome of interest as it seems more important than self-defined out-of-labor force status or reported disability status. For prime-aged men, we find that the changes in age, race, and ethnicity that occurred over this time period can "explain" 14 to 33 percent of the observed increase in full-year non-employment, without any change in disability insurance eligibility criteria or wages for low-skilled workers. For women, the observed changes in age, race, and ethnicity also would have led us to expect an increase in full-year non-employment, when in fact we saw dramatic decreases in non-employment.

Section II describes our data. Section III provides an overview of explanations for the increase in men not working. Section IV describes our empirical strategy and presents our results. Section V concludes.

II. Data

¹ Duggan and Imberman (2004) is a recent exception.

 $^{^{2}}$ We use 1979 through 2003 March CPS data throughout the analysis. The employment information corresponds to the previous year, thus 1978 to 2002. The demographic data (as well as some other information) corresponds to the survey year. However, for simplicity we refer to all data as coming from the year prior to the survey year, *i.e.*, the year corresponding to the employment information.

We use the March Current Population Surveys from 1979 to 2003 to examine changes in weeks worked over time.³ Questions about the labor market refer to the previous year. Our sample excludes people who may have not worked because of military service and those living in group-quarters. We examine weeks worked for men and women, aged 18 and over, but focus on those 30 to 50 years old. These are typically described as prime-aged workers, for whom changes over time in the likelihood of attending school and retirement behavior should not be driving their decisions about whether or not to work.

The number of weeks of non-employment is defined as 52 minus the number of weeks the individual reports working in the previous year. The number of weeks of nonparticipation is defined as 52 minus weeks worked last year and weeks spent looking for work or on layoff. Weeks unemployed is defined as weeks looking for work or on layoff.⁴ Percent of the year spent unemployed, out of the labor force, or non-employed is simply number of weeks spent in a given state, divided by 52.

Figures 1a, 1b, 2a, and 2b show the percent of year spent in unemployment, nonemployment, and nonparticipation, and percent of individuals reporting full-year nonemployment, for men aged 18 and over, women aged 18 and over, men aged 30 to 50, and women aged 30 to 50, respectively, from 1978 to 2003. Percent of the year spent in unemployment, non-employment and non-participation correspond to the measures of labor market (in)activity presented in Juhn, Murphy and Topel (2002) and percent of individuals reporting full-year non-employment corresponds to the measure we examine in detail below. Note that all of these measures show similar patterns. Figure 1a shows

³ We use the CPS data available through Unicon.

the pattern for men noted by Juhn, Murphy, and Topel (2002), namely, nonparticipation among men increased over the period. While unemployment rates were low by historical standards in the late 1990s, measures of not working—percent of year not participating, percent of year not employed, and percent of men with 52 weeks of non-employment were all higher than in previous periods. Figure 1b presents the same information for men aged 30 to 50. Again, while the percent of the year spent unemployed reaches historically low levels in the late 1990s, percent of the year spent in non-employment is no lower than it was in the late 1970s, and percent of year not participating and percent of men with full-year non-employment reach historically high levels in the late 1990s and early 2000s.

Figures 2a and 2b show these statistics for women aged 18 and over and women aged 30 to 50, respectively. Similar to men, women spent a very small percent of the year unemployed in the late 1990s. In contrast to men, however, all of the measures of non-employment show a dramatic decrease over the period, with a small increase in the most recent recession.

These figures suggest that some men have shifted from unemployment into nonparticipation in the labor market. The potential reasons for this shift are many, including changes in health that make work more difficult, increased participation in transfer programs that preclude work, and low demand for low-skilled workers that made job search less productive. Similar results presented by Juhn, Murphy, and Topel (2002) lead them to question whether the low unemployment rate during the 1990s really implied a robust labor market.

⁴ In the CPS weeks looking for work or on layoff applies to part-year workers. There is a separate question about weeks looking for work for nonworkers. We combine these to measure unemployment for everyone.

In contrast, women have shifted from nonparticipation into employment. The reasons for this change are similarly complicated although driven by different segments of the population. From 1978 to 2000 declines in labor force participation rates for men have been primarily concentrated among lower-skilled men as measured by level of completed education (See Anderson, Barrow, and Butcher 2004). Over this same period, labor force participation rates were increasing for women with at least a high school diploma and staying relatively constant among the women with less than a high school diploma.

III. Why Are Men Not Working?

Why are men, even prime-aged men, more likely to be in full-year nonemployment now than in the past? In addressing this question, researchers have focussed on changes over time in disability insurance criteria and other policies affecting the disabled, as well as changes in the labor market, particularly the labor market for lessskilled workers, that may have led some workers to leave the labor force. Below, we discuss three potential drivers of this trend. In addition to changes in disability policies and labor market structure, we consider how demographic changes may have contributed to trends in non-employment.

A. Changes in Policies Affecting the Disabled : 1979 to 2003^5

Disability insurance (DI) began with legislation signed by President Eisenhower in 1956. At the signing, Eisenhower clearly stated that DI was intended to increase

⁵ We draw on detailed descriptions in Stapleton et al. (1998), Social Security Administration (2001), and The Green Book (2000) for the following discussion.

economic security and "...to help rehabilitate the disabled so that they may return to useful employment...." During the 1970s, growth in DI enrollment and the recognition that few disabled people were ever leaving DI led Congress to enact legislation tightening eligibility requirements. Amendments to the Social Security Act between 1980 and 1981 introduced new work incentives for DI and SSI and required the Social Security Administration to tighten the way it adjudicated cases. The legislation limited the DI benefit level, required more careful screening of initial DI applicants, and required periodic review of participants' continuing disability status. To some extent, the 1980-1981 changes codified earlier administrative practices aimed at curbing the growth in DI receipt. However, the increased reliance on "continuous disability reviews" from 1982-1983 led to many people loosing their benefits. For example, termination rates (for medical reasons) per 1,000 beneficiaries reached a peak of 62 in 1982 compared to more typical rates of 5 to 25 per 1,000 beneficiaries (Autor and Duggan 2002). This led to a backlash against the tightening of eligibility requirements.

In response, 1984 legislation led instead to an expansion of the DI program. A moratorium was placed on continuing disability reviews. Impairment criteria reduced the importance of diagnostic or medical factors, with more weight given to functional factors such as the individual's limitations with respect to the activities of daily living. Further, "source evidence" from the individual's own physician had to be considered first, instead of relying heavily on the SSA consultative examination. Additionally, reviewers were allowed to consider the combined effect of multiple impairments in determining disability status. Thus, while an individual might not have any single impairment that would qualify her as disabled, she might qualify as disabled based on her group of

afflictions. Finally, benefits could no longer be terminated without proof of substantial medical improvements.

Changes in the laws governing disability insurance between 1984 and 1996 were relatively minor. However, some changes affected how the programs were administered. For example, in 1989, Congress mandated more SSI outreach. Court challenges have had a further effect. For example, the Sullivan vs. Zebley Supreme Court case increased the supply of benefits to children with mental disorders. Many speculate that this may have had spillovers to adults with mental impairments. Finally, downsizing at SSA in the early 1990s meant that there was less ability to review cases, and many believe that the adjudicative environment shifted in favor of making awards.

Eligibility standards were tightened somewhat in 1996. The Contract with America Advancement Act of 1996 excluded from eligibility persons whose drug or alcohol addiction was a primary factor in their disability, and it increased funding for reviews of the continuing disabled. Additionally, the Personal Responsibility and Work Opportunity Reconciliation Act of 1996 excluded many non-citizens from eligibility. The next major DI legislation occurred at the end of the Clinton administration with the introduction of the Ticket to Work and Work Incentives Improvement Act of 1999 which aimed to increase the incentives for the disabled to work.

To the extent that these legislative changes have bite, we would expect to see changes in disability rolls and corresponding changes in full-time, full-year nonemployment due to the loosening and expanding of eligibility requirements as described above. Changes in disability insurance applications, award rates, and beneficiaries are shown in figure 3. The shaded portions mark the NBER recession years. The tightening of DI rules in the early 1980s coincides with a decline in applications, awards, and worker beneficiaries as a share of the number of insured workers. In general, applications increase in recessions. Since the early 1980s, awards have increased from a low of about 3 per 1000 insured workers to the current high of nearly 6 per 1000 insured workers. The increase in award rates (a flow) has lead to a large increase in the number of disabled worker beneficiaries (a stock). This has increased from a low of about 25 beneficiaries per 1000 insured workers in the early 1980s to a high of about 41 beneficiaries per 1000 insured workers in 2003.

Recent research has examined the link between the rise in the disability insurance rolls and labor force status. Autor and Duggan (2003) find that the loosening of eligibility criteria in the 1980s and the increase in wage inequality (which had the effect of gradually raising the replacement rate of disability insurance) led to an increased propensity for high school dropouts to exit the labor force. According to their estimates, this may account for a one-half percentage point decline in the measured U.S. unemployment rate.

Bound and Waidmann (1992) examine the impact of disability transfers on labor force participation of older men. As an upper bound, they find that roughly one-half and one-third of the drop in labor force participation for 45 to 54 and 55 to 64-year-old men, respectively, can be linked to disability transfer programs. In a more recent paper, Bound and Waidmann (2002) find that *among the disabled*, growth in disability insurance receipt can account for most of the decrease in employment during the 1990s.

Other policy changes in the early 1990s may have changed the opportunities for disabled people to work. The Americans with Disabilities Act of 1990 required

employers to make reasonable accommodations for workers with disabilities. This requirement took effect for employers with 25 or more employees in July of 1992 and was extended to all employers of 15 or more employees in July of 1994 (Burkhauser and Daly (1996)). The intent of the law was to increase the ability of the disabled to find work; however, some researchers have found that the Act had the unintended consequence of reducing work opportunities for the disabled (DeLeire (2000a, 2000b) and Acemoglu and Angrist (2001)). Employers may have become more reluctant to hire disabled workers because they are now required to accommodate them in potentially expensive ways. The researchers argue that this decrease in demand for disabled workers contributed to their labor force withdrawal and an increase in the disability insurance rolls. If the ADA reduced work opportunities for the disabled, it may have correspondingly increased DI take-up rates and thus the percent of the population in fullyear non-employment. Alternatively, some research indicates that the disabled increased their school enrollment after the ADA which would also lead to an increase in full-year non-employment (Jolls 2004).

B. Business Cycle and Wage Structure Changes

Over the period of policy changes discussed above, the macro economy was also changing, going through both recessionary and expansionary periods. In addition, there were notable changes in the wage structure leading to increased wage inequality. (See, for example, Juhn, Murphy and Pierce (1993)).

Ceterus paribus, one would expect cyclical downturns to result in a lower percentage of people being employed for all 52 weeks of the year, *i.e.*, an increase in non-

employment for some number of weeks of the year. However, whether individuals describe themselves as unemployed or out of the labor force will depend on many factors, including alternate uses of their time, disability regulations, and the perceived benefits of job search. Juhn, Murphy, and Topel (2002) speculate that the shift from unemployment to nonparticipation for men is due both to (relative) improvements in their non-work opportunities, namely disability insurance payments, and to lower real wages for less-skilled workers.

In table 1 we present wage distribution information for five business cycle periods as defined by Juhn, Murphy, and Topel (2002) and for the most recent recession. From 1978 to 2002, there are three "peak" periods—1978-79, 1988-89, and 1999-2000—and three "trough" periods—1982-83, 1991-92, and 2001-2002. We show real wages at the 10th, 50th, and 90th percentiles of the distribution for men in the top panel. The same statistics are shown for women in the bottom panel.

As the top panel of table 1 shows, real wages for men at the 10th percentile of the distribution fell from the 1978-1979 peak through the 1991-1992 trough. By the 1999-2000 peak, real wages were on the rise, but were only about the same level as in 1978-1979. Wages at the median fell through the 1991-1992 trough as well, while wages at the 90th percentile rose throughout the periods shown.⁶ As discussed by Autor and Duggan (2002) and Juhn et al. (2002), declining wages at the lower end of the skill distribution may particularly have affected disability insurance take-up. As wages for less-skilled workers were falling, disability payments became more valuable relative to the wages less-skilled men would likely receive.

Women similarly experienced declines in real wages at the 10th percentile. However, wages at the 10th percentile begin to rise sooner than men's wages at the 10th percentile, and at the median women's wages are rising throughout.

C. Changes in Health and Demographic Change

The literature examining the increase in men's non-employment focuses on the role of changes in the regulatory environment affecting disability and changes in the labor market that may have reduced the demand for low-skilled workers in relatively ill health. Less research in economics has addressed changes in health itself as a potential cause of the increase in the disability rolls and non-employment.⁷ Many of the papers on the increase in disability rolls mention that this increase happened against a backdrop of increasing aggregate health, typically measured as decreasing mortality. However, researchers in public health have focussed their attention on inequalities in health outcomes. Morbidity and mortality are inversely related to income (and to many other measures of socio-economic status—see Deaton 2002 for a recent review). Thus, it is possible that aggregate health may have been increasing at the same time as health inequality was growing, just as average real wages may increase while real wages of those at the bottom of the wage distribution are stagnant or falling.⁸

Recent evidence suggests that despite declining mortality rates, morbidity is increasing. Lakdawalla, Bhattacharya, and Goldman (2004) define disability by survey

⁶ Wages at the 10th percentile may be driven, in part, by the real value of the minimum wage. For example, the federal minimum wage (in 2003 dollars) was \$6.38 in 1979, fell to \$5.56 in 1983, fell further to \$4.51 in 1989, and then rose to \$5.13 in 1992.

⁷ Cutler and Richardson (1997) describe changes in health in the United States. The 20th Century saw a dramatic decrease in mortality, although morbidity changes are less clear.

⁸ Many public health researchers go further and posit a direct causal impact of inequality on health. This would imply that rising wage inequality leads to worse health (see, for example, Wilkinson 1996).

questions about personal care limitations. Using this definition of disability, they find that from 1984 to 1996, disability among the non-elderly rose. The largest increase in disability was for those aged 30-39 for whom disability rose more than 50 percent. In part, this increase in disability among the non-elderly may be driven by steep increases in obesity (see Anderson, Butcher, and Levine 2003 for an overview). It may also be the case that morbidity has increased precisely because mortality has decreased, i.e., those who would have died in the past now survive, but are disabled.

In addition to the evidence that that health has declined across population groups, we know that there have been increases in population groups that tend to have higher morbidity. For a simple example, the population is aging, and age is related to increased morbidity. Thus if health of the working-aged population is declining, we would expect to see an increase in the disability rolls and an increase in full-year non-employment even without changes in disability insurance criteria.

Here, we briefly review the demographic changes that are the focus of our study and the link between these changes and health. We mainly focus on changes in the age structure of the population and changes in race and ethnicity. In addition, we briefly examine the role of changes in marital status and education for non-employment. In Table 2 we show how age, race, ethnicity, marital status, and education have changed for the adult population over the six time periods we examine. For men, median age increased from 39 in 1978-1979 to 43 in 2001-2002; similarly, median age increased from 41 to 44 for women.⁹ In figure 4, we plot the age distributions for individuals aged 18 to 80 from the 1979 and 2003 March CPSs. The aging of the baby-boom cohort from their twenties into their forties is seen in the rightward shift of the mass of the distribution.

Because morbidity increases with age, the aging of the population may have an important effect on disability and non-employment. The National Center for Health Statistics (NCHS) publishes detailed statistics on how health measures change with certain demographic characteristics (Blackwell and Tonthat 2002).¹⁰ The NCHS summary of health statistics for the United States shows that the percent of the population reporting their health as "fair" or "poor" increases from 5.3 percent for those aged 18 to 44 years to 14 percent for those aged 45 to 64 years. The percent of the population with any activity limitation increases from 6.7 to 17.5 percent for these age groups.¹¹

In addition, the percent of the population that is white declined from 88.6 to 83.5 percent for men and from 87.3 to 81.6 percent for women between 1978-1979 and 2001-2002. Both the percent of the population that is African American and the percent that is Hispanic have increased.¹² The NCHS reports self-assessed health is worse for African Americans (non-Hispanic) and Hispanics than it is for white non-Hispanics.¹³ Roughly 8.2 percent of white non-Hispanics assess their health as "fair" or "poor" while12.9 percent of African Americans fall into these categories. This statistic for Hispanics is 8.8 percent. Activity limitations also differ along racial and ethnic lines, with 13.4 percent of white non-Hispanics and 14.7 percent of blacks reporting some activity limitation.

⁹ Beginning in 2002 age is topcoded at 80 years in the March CPS. For consistency over time we topcode age at 80 years for all CPS samples. As a result, average age of the population is understated in each period.
¹⁰ These statistics come from NCHS analyses of the 1998 National Health Interview Surveys.

¹¹ See NCHS tables 2 and 4.

¹²Note that these calculations are based on the Current Population Surveys that sample the civilian noninstitutionalized population. As others have noted (see for example Katz and Krueger (1999) and Chandra (2003)), the increase in the prison population over this period disproportionately affected African-American and Hispanic men, and thus, the CPS will understate demographic change overall since these men are not in the sampled population.

Despite lower health assessments, Hispanics report a lower percentage of activity limitations than whites or African Americans (8.4 percent).

The percent of the population that is married has declined over the period. For men, the percent who are never married increases from 18.6 to 23.7 from the beginning to end of our sample period, and the percent who are divorced increases from 5.3 to 9.3. Similarly, the percent of women never married increases from 12.8 to 17.9, and the percent divorced increase from 7.2 to 11.5. Marriage may provide financial, emotional, and in-kind support that may directly affect health, or may affect an individual's take-up of disability transfer programs for a given level of health (Stapleton et al. 1998). In addition, mental illness, one of the forms of disability that has increased in recent years, is negatively related to marriage (Bartel and Taumban 1986). Finally, whether one returns to work following a period of disability is also correlated with marital status. Married men are more likely to return to work than unmarried men; conversely unmarried women are more likely to return to work than married women (Baldwin and Johnson 2001).

Finally, the population has become more educated. For men the numbers without at least a high school diploma have declined from 30.2 percent in 1978-79 to 17.4 percent in 2001-02 while the numbers with at least some college have increased from 39 percent to 51.5 percent. The education levels of women have increased similarly.

Clearly, it is difficult to discern the direction of causality when it comes to the relationship between either marital status or education and health and disability. People who have worse health are, perhaps, less successful in the marriage market. On the other hand, marriage may have an impact on one's health. Further, for a given level of health,

¹³ See NCHS tables 2 and 4.

being married may facilitate work, or, alternatively, may provide resources such that work is less necessary. Similarly, researchers have found that higher levels of education are correlated with better health. Thus, we mainly focus on changes in age, race, and ethnicity as these changes are the most likely to be exogenous.

In what follows, we consider changes in non-employment over time for men and women aged 18 and over, with a particular focus on prime-aged individuals, 30 to 50 years old. We focus on non-employment, rather than non-participation, because it does not depend on individuals' self-definition of their activities and is therefore more easily compared across time periods. We try to disentangle the effects on non-employment of changes in population characteristics that may be linked to health, from changes in the labor market, the business cycle, and the regulatory environment governing disability.

IV. Empirical Strategy and Results

A. Empirical Techniques

In order to examine the role that changes in age, race, and ethnicity play in the changes in full-year non-employment for men and women, we have adapted techniques developed by DiNardo, Fortin, and Lemieux (1996). They examine questions, such as: How would the wage distribution have changed from 1973 to 1992 if union participation had remained at its 1973 level? We adapt their techniques to examine the distribution of weeks worked (or not worked). We want to ask, for example, how would the distribution of weeks not worked have changed from 1978 to 2000 if the regulatory and macroeconomic environments had remained as they were in 1978, but age, race, and ethnicity changed to their 2000 levels?

Consider the following distribution of weeks worked:

$$g(w) = \int f(w \mid x)h(x)dx \tag{1}$$

where f(w | x) is the density of weeks worked (w) conditional on a set of characteristics (x). The set of characteristics x has distribution $h(\cdot)$.

The observed density of weeks worked in 2000 is:

$$g(w \mid t = 2000) = \int f^{2000}(w \mid x)h(x \mid t = 2000)dx$$
(2)

Similarly, the observed density of weeks worked in 1978 can be written as:

$$g(w \mid t = 1978) = \int f^{1978}(w \mid x)h(x \mid t = 1978)dx$$
(3)

We would like to know what the distribution of weeks worked would look like if the population had the distribution of characteristics as in 2000, but the 1978 distribution of weeks worked conditional on those characteristics. This can be written as:

$$g_{2000}^{1978}(w) = \int f^{1978}(w \mid x)h(x \mid t = 2000)dx$$
(4)

Bayes Law implies that:

$$h(x) = \frac{h(x \mid t = 1978) \operatorname{Pr}(t = 1978)}{\operatorname{Pr}(t = 1978 \mid x)}$$

and

$$h(x) = \frac{h(x \mid t = 2000) \operatorname{Pr}(t = 2000)}{\operatorname{Pr}(t = 2000 \mid x)}$$

where Pr(t = 1978) and Pr(t = 2000) are the probabilities that a given sample comes from 1978 and 2000. Pr(t = 2000 | x) and Pr(t = 1978 | x) are the probabilities that a sample comes from a particular year, given the observed characteristics. We can set

Pr(t = 1978) = Pr(t = 2000)—the probability of a sample coming from 2000 and 1978 is the same—so we can re-write the distribution of characteristics in 2000 in terms of the distribution of characteristics in 1978 and the probabilities that a given sample comes from a given year, given the observed characteristics as in (5):

$$h(x \mid t = 2000) = h(x \mid t = 1978) \cdot \frac{\Pr(t = 2000 \mid x)}{\Pr(t = 1978 \mid x)}$$
(5)

Thus, we can re-write what the distribution of weeks worked in 1978 would have been, if 1978 had had the same demographic characteristics as 2000:

$$g_{2000}^{1978}(w) = \int f^{1978}(w \mid x)h(x \mid t = 2000)dx = \int \theta(x)f^{1978}(w \mid x)h(x \mid t = 1978)dx$$
(6)

where $\theta(x) = \frac{\Pr(t = 2000 \mid x)}{\Pr(t = 1978 \mid x)}$.

In other words, we have reduced the problem to one of reweighting the 1978 data with the ratio of the probability that an observation comes from 2000, given observed characteristics, to the probability that an observation comes from 1978, given observed characteristics. We call θ the "counterfactual" weights.

Consider reweighting the 1978 data to have the same age distribution as in 2000. We can take the 1978 data and create "counterfactual" weights such that the weighted age distribution of the 1978 data is precisely the same as the age distribution in 2000. This means creating weights that increase the weight of older people in the 1978 data and decrease the weight of younger people. We can then consider reweighting the distribution of weeks worked in 1978 with the weights described above. This is equivalent to asking what the weeks worked distribution was in 1978 among people with the same age distribution as in 2000. By using the 1978 data, we are ensuring that these people with the 2000 age distribution are "facing" the regulatory and macro environments that existed in 1978.

To construct the counterfactual weights, we first normalize the CPS weights to sum to one in each year. Then, using two years of data, we estimate a logistic model using the CPS normalized weights to predict from which year an observation comes as a function of demographic characteristics.¹⁴ We then multiply the normalized CPS weight by the appropriate estimated odds ratio from the logistic regression. This is our counterfactual weight.

B. Results

i. Results for 1978 and 2000

Before turning to the results of this counterfactual weighting strategy, consider tables 3a and 3b. These tables give us some insight into how changes in the age distribution may affect the distribution of weeks worked. The first two columns show the percent of the population in 13 age categories in 1979 and 2001. From 1979 to 2001 about 30 percent of the adult male population (table 3a) moves from being in the three youngest age categories to being in the three age categories spanning 35 to 49. This is the aging of the baby boom. Labor economists normally think of this aging of the population as increasing the number of weeks worked as these are the years when workers are at the

¹⁴ The included demographic characteristics are as follows: a fifth-order polynomial in age, indicators for race is African American and race is other non-white, an indicator for Hispanic ethnicity, and all pair-wise interactions. The results with marital status additionally include indicators for all marital status categories

peak of their age-earnings profile. Indeed, if we look at the next two columns, the fraction of the population working 52 weeks per year, we can see that these 3 age categories have the highest percent of workers working full-year. However, if we look at the last two columns, these show the percent of the population working no weeks per year, or full-year non-employed. In 1979, that fraction starts to increase with the 40 to 44 year-old category, and increases with each age group thereafter. The patterns are similar in 2001. The age patterns are similar for women, although between 1979 and 2001 we see the large increase in women's labor supply.

The re-weighting technique that we described above allows us to examine how the distribution of weeks worked has changed over time in a more straightforward fashion. In figure 5 we present a bar graph for changes in weeks not worked for men aged 18 and over between 1978 and 2000. We create 7 categories for weeks not worked: zero, 1 to 12, 13, 14 to 25, 26, 27 to 51, and 52 weeks. The "zero" category represents those who worked all weeks in a given year. We break out 13 and 26 weeks separately, since these represent small mass points in the full distribution of weeks worked, likely related to unemployment insurance rules.

The first series in the graph represents the observed change from 1978 to 2000 in the percent of men who did not work for that particular number of weeks. For example, 56.8 percent of men worked for the full 52 weeks in 1978 and 62.2 percent worked 52 weeks of the year in 2000. The first column in the figure 5 bar chart represents this 5.4 percentage point change in the number of men working 52 weeks of the year. At the other end of the spectrum, 17.1 percent of men did no work for 52 weeks in 1978 and 21.6

⁽married spouse present is the omitted category) and pair-wise interactions with the age, race, and ethnicity variables above.

percent of men did no work for 52 weeks in 2000, an increase of 4.5 percent. Thus, the first series represents the actual change in the distribution of weeks worked from 1978 to 2000. From 1978 to 2000 there were increases in both the percent of men working 52 weeks of the year and the percent of men working no weeks in the year. Looking at the other categories, there were relatively large decreases in the fraction of men who did not work 1 to 12 weeks or 27 to 51 weeks. Compared to 1978, it looks as though weeks worked shifted to the extremes of the distribution in 2000. Men who had not worked for a small number of weeks in previous years may have shifted into full year employment, and the men who had not worked for quite a few weeks in earlier years may have shifted into full year non-employment.

The second series (diagonal lines) represents the "counterfactual" change in the distribution of weeks not worked from 1978 to 2000 for men aged 18 and over. The counterfactual change equals the predicted distribution of weeks not worked in 2000 minus the observed distribution of weeks not worked in 1978. Our prediction of weeks not worked in 2000 reweights the 1978 data to have the same age distribution as the 2000 men. In other words, this series represents the change in the distribution of weeks not worked if we observed the actual change in age, but under the 1978 "environment." By "environment" we mean, what if the men in 2000 had faced the regulatory and macroeconomic conditions that existed in 1978.¹⁵

We see that the changes in demographic characteristics would have led us to predict an increase in the fraction of the population that did not work for the full 52 weeks, even if there were no change in the policy and macroeconomic "environment." In fact, changes in the age distribution "explain" nearly 14 percent of the observed change. Under this identification strategy, the remaining difference can be attributed to changes in the regulatory and macroeconomic environments. Age changes predict a somewhat larger increase in the percent of men working 52 weeks of the year.

Figure 6 presents the same exercise for men aged 30 to 50. Here we are asking how the change in the age distribution among the prime aged affected the distribution of weeks worked. Again we see that the changing distribution of age among the prime aged would have predicted an increase in both full-year employment and full-year nonemployment, although a smaller amount of the change is explained when we focus on prime aged men.

 Changes in Demographic Characteristics versus Changes in Regulatory and Macroeconomic Conditions

In what follows, we focus exclusively on prime aged men and women. We also limit our analysis to the fraction of people full-year non-employed, instead of examining the entire distribution of weeks worked. The increase in full-year non-employment for prime age men poses the biggest puzzle and the greatest concern since if they are on now on disability insurance, they are likely to be on it for a long time. We consider the effect of the changes in four demographic characteristics, age, race and ethnicity, marital status, and education. We also examine the effect of these changes in characteristics under different regulatory and macroeconomic environments.

In tables 4a-7b, we perform the same sort of counterfactual exercise as above for men and women but limit our analysis to full-year non-employment for prime-aged men

¹⁵ Here we generate our counterfactual weights using a fifth-order polynomial in age.

and women. We also break the years into peak and trough years as in tables 1 and 2. Comparing periods in which the macroeconomic conditions are similar, for example, comparing peak periods to peak periods, we can assess the importance of macroeconomic and regulatory changes versus demographic changes.

In tables 4a-7b the numbers on the diagonal present the actual data. For example, in table 4a, in 1978-79, 4.17 percent of men were not employed for the full year; in 2001-2002, 8.19 percent of men were not employed for the full year. The general trend is upward over each of the periods although the percentage jumps up during trough years and dips down during peak years.

The off-diagonal elements of the tables correspond to counterfactual distributions of weeks not worked calculated as described in the previous section. For example, row five of column 1 in table 4a represents the 1978-79 distribution of weeks not worked reweighted by the 1999-2000 age distribution. Similarly, the first row of the fifth column represents the 1999-2000 (period 5) distribution of weeks not worked reweighted by the 1978-79 (period 1) age distribution. Following column 1 from row 1 to row 6 we are asking, "How would people who existed in each of the periods have behaved if they had faced the environment that prevailed in the 1978-79 period?" The other columns correspond to the same exercise, but using a different period for the "environment." For example, column 4 takes the 1991-92 period as the "environment" and asks, "How would people who existed in each of the periods have behaved if they had faced the environment that prevailed in the 1991-92 period?" Tables 4a-7b present these results for men (the "a" tables) and women (the "b" tables), each using a different set of demographic characteristics. Tables 4a and 4b reweight by age only; tables 5a and 5b

reweight by age, race, and ethnicity; tables 6a and 6b reweight by age, race and ethnicity, marital status; and tables 7a and 7b reweight by age, race and ethnicity, marital status, and education.

What do these tables tell us? First, each of the columns traces out the effect of demographic change under different regulatory and macroeconomic conditions. Consider table 5a. The fact that the share of men who spent the full year non-employed increases as we look down each column suggests that demographic change would have led to an increase in the fraction of men who spent the full year non-employed, regardless of the regulatory and macroeconomic environments they faced. In other words, regardless of the environment, older, nonwhite, and Hispanic men are more likely to be full-year non-employed than younger, white, non-Hispanic men. The fact that men are aging, even within the prime age group, and are more likely to be African American and Hispanic has led to an increase in the fraction of men who are full-year non-employed.

Second, the table can give us some insight into the relative importance of changes in the regulatory and the macroeconomic environments. Note that the column 1, 3, and 5 counterfactual estimates correspond to business cycle peaks (1978-79, 1988-89, and 1999-2000, respectively) and within each row these estimates are less than the counterfactual estimates that correspond to trough periods (1982-82, 1991-92, 2001-02). Clearly, macroeconomic fluctuations are important in determining the fraction of men that is not working.

We can estimate the predicted increase in the percent of men with full-year nonemployment between periods in two ways. For example, the predicted increase in the percent of men with full-year non-employment between periods 1 and 5 can be calculated as: (1) the difference between the predicted percent not working 52 weeks in 1999-2000 under the 1978-79 environment (column 1, row 5) minus the observed percent of men not working 52 weeks in 1978-79 (column1, row 1), or (2) the difference between the observed percent not working 52 weeks in 1999-2000 (column 5, row 5) minus the predicted percent not working 52 weeks in 1978-79 under the 1999-2000 environment (column 5, row 1). As in the Oaxaca/Blinder (1973) decomposition of wage differences by sex where it matters whether one uses the coefficient estimates for men or the coefficient estimates for women in calculating the shares of the observed wage difference due to differences in human capital, the predicted change in full-year non-employment may depend on which combination of demographic weights and "environment" we use.

In table 8a we present the predicted changes in full-year non-employment and shares of the actual change attributable to demographic changes for a select subset of time periods. We include one peak-to-peak and one trough-to-trough comparison. We chose these particular time periods because they represent the longest time spans. We show the effect for age alone; age, race and ethnicity; age, race and ethnicity and marital status; and age, race and ethnicity, marital status, and education. In each case we calculate the predicted change using both of the re-weighting schemes described above.

In order to think about the two methods of decomposing the change in percent not working 52 weeks of the year, consider a simple example in which there are only two types of men, young and old, and two time periods, 1978 and 2000. Further assume that the probability of being old given that the year is 1978 equals p_1 , and the probability of being old given that the year is 2000 equals p_2 . The observed share of men experiencing full year non-employment (y) in 1978 can be written as

$$(1-p_1)\Pr^{1978}(y \mid young) + p_1\Pr^{1978}(y \mid old).$$
 (7)

The counterfactual share of men experiencing full year non-employment in 2000 assuming the conditional probabilities of not working full year are the same as in 1978 can be written as

$$(1-p_2)\Pr^{1978}(y \mid young) + p_2\Pr^{1978}(y \mid old).$$
 (8)

Then the predicted change in the share of men in full-year non-employment equals

$$(p_1 - p_2) \left[\Pr^{1978}(y \mid young) - \Pr^{1978}(y \mid old) \right]$$
(9)

Similarly, the predicted change in the share of men in full-year non-employment, assuming the conditional probabilities of full-year non-employment equal those observed in 2000 can be written as

$$(p_1 - p_2) \left[\Pr^{2000}(y \mid young) - \Pr^{2000}(y \mid old) \right]$$
(10)

Consider equations (9) and (10) in a world in which the population is aging, in other words, $p_2 > p_1$. As long as the probability of full-year non-employment for the old exceeds the probability of full-year non-employment for the young in both years, we will predict an increase in the share of men with full-year non-employment. The size of the predicted change may be different if the difference in the probabilities of full-year non-employment between the young and old change over time. If the probability of full-year non-employment conditional on being old is less than the probability of full-year non-employment conditional on being young in the later period, the predicted change in full-year non-employment will be positive using equation (9) and negative using equation (10). This example demonstrates how using different base years for the "environment" under which we evaluate the impact of demographic change can generate different predictions.

With that simple example in mind, consider table 8a. The top panel shows that between 1978-79 and 1999-2000, there was a 2.94 percentage point increase in the number of men who were full-year non-employed.¹⁶ The top row shows that using the 1978-79 "environment," the change in the age distribution would have led to a 0.11 percentage point increase in full-year non-employment. Using the 1999-2000 environment, we would have predicted a 0.16 percentage point increase in full-year non-employment. The change in the age distribution alone can explain 3.7 percent of the increase in full-year non-employment under the earlier macro economic and regulatory conditions. The change in the age distribution can explain about 5.4 percent of the increase under the later conditions.

Including race and ethnicity leads to a larger predicted increase in full-year nonwork, under either set of macro economic or regulatory conditions. The changes in these demographic characteristics account for 13.9 to 18.4 percent of the increase in full-year non-employment. The bottom panel presents these comparisons for two trough periods, 1982-83 and 2001-02. Here, age, race and ethnicity changes can account for about onethird of the increase in full-year non-employment, using either period as the "environment."

The fact that we find similar predicted increases in non-employment using the regulatory/macroeconomic conditions of either 1978-79 or 1999-2000 and 1982-83 or 2001-02 as the "environment" means that under any of these regulatory or macroeconomic conditions, the observed changes in age, race and ethnicity, lead to increases in the fraction of men who are full-year non-employed. This suggests that

¹⁶ The difference in this calculation from the calculation applying to figure 6 arises because we are considering 1978-79 and 1999-2000 instead of single year comparisons.

older non-whites and/or older Hispanics were always more likely to be full-year nonemployed, and in the later period there are more of them.

There have been many changes in the population since the late 1970s. We choose to focus on age and ethnicity because these changes are the most clearly exogenous. However, one can employ this technique to assess the impact of any changes in population characteristics. In the last two rows of each panel of table 8a, we include marital status or marital status and education. As discussed above, there is a strong correlation between health, marital status, and education, and a correlation between full-year non-employment, marital status, and education. Adding marital status to the list of demographic characteristics increases the proportion of the change in full-year non-employment that can be explained by changes in demographic characteristics. In the sector of the increase in full-year non-employment is explained by the changes in these characteristics; in the trough-to-trough comparison over 100 percent of the increase in full-year non-employment is explained by these changes.

The last row of each panel in table 8a adds education to the list of demographic variables. For 1978-79, the changes in demographic characteristics that occurred between 1978-79 and 1999-2000 would predict about 17 percent of the actual increase in non-employment that occurred between those two periods. For the trough-to-trough comparison, we predict about 43 percent of the increase using the 1982-83 environment and 27 percent of the increase using the 2001-2002 environment. For 1999-2000 environment, the demographic changes between 1978-79 and 1999-2000 predict a decline in full-year non-employment, thus, we explain a negative percent of the actual

change. As in the simple example above, the fact that the sign of the prediction changes between the two periods suggests that either the change in macro economic and regulatory environment disproportionately affects individuals of some types, or that it does not mean the same thing to be an individual of a given type in the two periods. For example, the average unobserved "ability" of a high school dropout in 1978-79 may be higher than the average unobserved "ability" of a high school dropout in 1999-2000. One can think of similar arguments about the average ability among divorced men in the earlier and later periods. On net, changes in education seem to dominate when we use 1999-2000 as the base period.

The results for women, in table 8b, are quite different. The percentage of women who were full-year non-employed fell between the peak periods by 12 percentage points and between the trough periods by 7.5 percentage points. Demographic characteristics changed for women in the same ways that they did for men: namely, the population of women got older and are less likely to be white, non-Hispanic, and married. Women's education also increased.

Consider the changes in age, and age, race and ethnicity in the top two rows of each panel. For women, these changes in characteristics either explain a very small proportion of the overall decrease in full-year non-employment or predict an increase. For women, the women's liberation movement, changes in home production technology, and welfare reform may be much more important in determining women's full-time nonemployment than changes in the demographic characteristics considered thus far. See Greenwood *et al.* (2002) for example. When we include marital status and education, we do much better at predicting the actual change in full-year non-employment. We predict between 31 and 44 percent of the peak-to peak or trough-to-trough decline in full-year non-employment regardless of which period's macro economic and regulatory environment we use to examine the effect of demographic change.

V. Conclusion

Previous literature examining the increase in men's non-employment has focused on the roles of disability policies and macroeconomic conditions. In this analysis, we examine how demographic changes between 1978 and 2003, in particular, age, race, and ethnicity, may also contribute to the increase in non-employment for men 30 to 50 years old. These demographic characteristics are correlated with health and disability status; therefore, even without the changes in disability policies in 1984 and 1990, we would have expected an increase in full-year non-employment. Under our maintained assumptions, this analysis shows that about 15 to 30 percent of the increase in men's fullyear non-employment may be explained by demographic changes alone.

The counterfactual analysis for women is similar to that for men in that changes in age, race, and ethnicity over the period are predicted to increase the fraction of women who are full-year non-employed. In contrast to men, however, women's actual nonemployment fell over the period. Women increased their employment, even in the face of underlying demographic and policy changes that would be expected to decrease it.

References

- Acemoglu, Daron and Joshua Angrist, "Consequences of Employment Protection: The Case of the Americans with Disabilities Act," *Journal of Political Economy*, 2001, 109:915-57.
- Anderson, Patricia, Kristin F. Butcher, and Phillip Levine "Economic Perspectives on Childhood Obesity," *Federal Reserve Bank of Chicago Economic Perspectives*, Third Quarter 2003, vol. 27, pp. 30-48.
- Anderson, Katharine, Lisa Barrow, and Kristin F. Butcher, "Implications of Changes in Men's and Women's Labor Force Participation for Real Compensation Growth, Inflation, and the Natural Rate," forthcoming in *Topics in Economic Analysis & Policy*.
- Autor, David H., and Mark G. Duggan, "The Rise in the Disability Rolls and the Decline in Unemployment," *Quarterly Journal of Economics*, February 2003, v. 118, no. 1, pp.157-206.
- Baldwin, Marjorie L., and William G. Johnson, "Dispelling the Myths about Work Disability," prepared for the 1998 IRRA Research volume *New Approaches to Disability in the Workplace*, mimeo, February 1, 2001.
- Bartel, A. and P. Taubman, "Some Economic and Demographic Consequences of Mental Illness," *Journal of Labor Economics*, 1986, vol. 4, no. 2, pp. 243-256.
- Blackwell, Debra, and Luong Tonthat, "Summary Health Statistics for the U.S. Population: National Health Interview Survey 1998," National Center for Health Statistics, Vital Health Stat 10(207), 2002.
- Blinder, Alan S., "Wage Discrimination: Reduced Form and Structural Estimates" *Journal of Human Resources*, Fall 1973, v. 8, iss. 4, pp. 436-55.
- Bound, John, and Timothy Waidmann, "Disability Transfers, Self-Reported Health, and The Labor Force Attachment of Older Men: Evidence from the Historical Record," *Quarterly Journal of Economics*, November 1992, p. 1293-1419.
- Bound, John, and Timothy Waidmann, "Accounting for Recent Declines in Employment Rates Among Working-Aged Men and Women with Disabilities," *Journal of Human Resources*, Spring 2002, v. 37, iss. 2, pp. 231-50.
- Chandra, Amitabh, "Is the Convergence in the Racial Gap Illusory?," NBER working paper No. 9476, January 2003

- Cutler, David and Elizabeth Richardson, "Measuring the Health of the U.S. Population," *Brookings Papers on Economic Activity*, 1997, p.217-271.
- Deaton, Angus, "Policy Implications of the Gradient of Health and Wealth," *Health Affairs*, March/April 2002, p. 13-30.
- DeLeire, Thomas, "The Unintended Consequences of the Americans with Disabilities Act," *Regulation*, 2000a, 23(1):21-24.
- DeLeire, Thomas, "The Wage and Employment Effects of the Americans with Disabilities Act," *Journal of Human Resources*, 2000b, 35(4): 693-715.
- DiNardo, John, Nicole Fortin, and Thomas Lemieux, "Labor Market Institutions and the Distribution of Wages, 1973-1992: A Semi-Parametric Approach," *Econometrica*, 1996, vol. 64, no. 5 (September), pp. 1001-44.
- Duggan, Mark and Scott Imberman, "Why Are the DI Rolls Skyrocketing? The Contribution of Population Characteristics, Economic Conditions, and Program Changes," mimeo, University of Maryland, September 2004.
- Greenwood, Jeremy, Ananth Seshadri, and Mehmet Yorukoglu. "Engines of Liberation," Unpublished manuscript, University of Rochester. 2002.
- House Ways and Means Committee, 2000 Green Book, 106th Congress. Government Printing Office: Washington, D.C., 2000.
- Jolls, Christine, "Identifying the Effects of the Americans with Disabilities Act Using State-Law Variation: Preliminary Evidence on Educational Participation Effects," May 2004. National Bureau of Economics Working Paper No. 10528
- Juhn, Chinhui, Kevin M. Murphy, and Brooks Pierce, "Wage Inequality and the Rise in Returns to Skill," *Journal of Political Economy*, June 2003. v. 101, issue 3, pp. 410-42.
- Juhn, Chinhui, Kevin M. Murphy, and Robert H. Topel, "Current Unemployment, Historically Contemplated," *Brookings Papers on Economic Activity*, I:2002, pages 79-135.
- Katz, Lawrence and Alan Krueger, "The High-Pressure U.S. Labor Market of the 1990s," Brookings Papers on Economic Activity, 1999. v. 0, issue 1, pp. 1-65.
- Lakdawalla, Darius, Jayanta Bhattacharya, and Dana P. Goldman, "Are the Young Becoming More Disabled? Rates of Disability Appear to be on the Rise Among People Ages Eighteen to Fifty-nine, Fueled by a Growing Obesity Epidemic," *Health Affairs*, January/February 2004 v. 23: 168-176

- Oaxaca, Ronald, "Male-Female Wage Differentials in Urban Labor Markets," International Economic Review, October 1973, v. 14, iss. 3, pp. 693-709.
- Social Security Administration, Annual Report on the Social Security Disability Insurance Program, 2001. Office of Policy. Office of Research, Evaluation, and Statistics. 2001.
- Stapleton, David, Kevin Coleman, Kimberly Deitrich, and Gina Livermore, "Empirical Analyses of DI and SSI Application and Award Growth," in *Growth in Disability Benefits: Explanations and Policy Implications*, Kalman Rupp and David C. Stapleton, editors, 1998, W.E. Upjohn Institute for Employment Research: Kalamazoo, MI, p. 63-76.
- Wilkinson, Richard, Unhealthy Societies: The Afflictions of Inequality, Routledge: London, 1996.













Figure 2b: Percent of Weeks per Year Spent in Unemployment, Nonemployment, and Nonparticipation and Percent of Women with 52 Weeks of Nonemployment, Women Ages 30 to 50





Figure 3: Disability Insurance Applications, Awards, and Beneficiaries per 1000 Insured Workers



Distribution of the Population by Age 1979 and 2003

Notes: Estimates are based on the population ages 18 to 80 years old and use the Epanechnikov kernel.



Figure 5: Observed and Predicted Changes in the Distribution of Weeks Not Employmed, Men Ages 18 and Over

Notes: The predicted change in the distribution of weeks not employed equals the difference between the predicted distribution in 2000 and the observed distribution in 1978. The predicted 2000 distribution is calculated using the 1978 distribution of weeks not employed reweighted by the 2000 age distribution.



Figure 6: Observed and Predicted Changes in the Distribution of Weeks Not Employed, Men Ages 30 to 50

	10th Percentile	Median	90th Percentile
		Men	
Peak 78-79	6.19	15.09	29.10
Trough 82-83	5.66	14.44	29.66
Peak 88-89	5.50	14.50	31.15
Trough 91-92	5.47	14.19	30.73
Peak 99-00	6.24	15.60	36.91
Trough 01-02	6.46	15.90	39.18
		Women	
Peak 78-79	4.57	9.20	17.75
Trough 82-83	4.45	9.34	18.78
Peak 88-89	4.44	10.21	21.47
Trough 91-92	4.62	10.34	22.23
Peak 99-00	5.20	11.96	26.80
Trough 01-02	5.47	12.42	28.42

Table 1Hourly Wage Distribution by Business Cycle Period

Notes: Distributions are for adults 18 years of age and older based on authors' calculations from the March Current Population surveys. See text for additional sample restrictions. All wages are reported in real 2003 dollar based on the Personal Consumption Expenditures, chain-type price index.

Demograp	nic Characteris	stics of Persons	Aged 18 and 0	Over by Busines	s Cycle Period	
	Peak	Trough	Peak	Trough	Peak	Trough
	78-79	82-83	88-89	91-92	99-00	01-02
			Ν	1en		
Age in years						
Mean	42.2	42.1	42.7	43.2	44.0	44.3
Median	39	39	39	40	42	43
Race/Ethnicity						
White	88.6	87.5	86.3	85.7	84.1	83.5
African-American	9.6	10.0	10.4	10.6	11.1	10.6
Hispanic	4.9	5.8	7.6	8.2	10.9	12.7
Marital Status						
Married spouse						
present	66.2	63.7	61.3	60.2	57.9	57.2
Married spouse						
absent	0.7	0.8	0.8	0.9	1.5	1.6
Widowed	2.7	2.6	2.7	2.8	2.7	2.6
Divorced	5.0	6.0	7.2	7.6	8.8	8.7
Separated	2.0	2.0	2.1	2.1	1.7	1.8
Never married	23.3	25.1	26.0	26.4	27.3	28.1
Education						
Elementary	14.2	12.0	9.6	8.8	6.5	6.5
High school	16.0	14.6	13.4	11.8	10.7	10.9
HS graduate	31.0	32.1	32.7	33.2	31.8	31.1
Some college	21.0	21.4	22.4	24.1	26.1	25.8
College graduate	17.9	20.0	21.8	22.1	24.9	25.7
			Wo	omen		
Age in years						
Mean	43.7	43.8	44.4	44.9	45.6	46.0
Median	41	40	41	42	43	44
Race/Ethnicity						
White	87.3	86.1	85.0	84.4	82.5	81.6
African-American	10.9	11.3	11.7	12.0	12.7	12.4
Hispanic	4.7	5.7	7.1	7.6	10.3	11.3
Marital Status						
Married spouse						
present	59.7	57.5	56.2	55.5	53.7	53.2
Married spouse						
absent	0.8	0.7	0.7	0.7	1.3	1.3
Widowed	12.9	12.4	12.2	11.6	10.6	10.4
Divorced	6.9	8.1	9.2	10.0	10.9	11.4
Separated	3.0	3.0	2.9	3.0	2.5	2.6
Never married	16.8	18.2	18.8	19.1	21.1	21.2
Education						
Elementary	13.8	11.6	9.2	8.5	6.2	6.0
High school	16.6	15.0	13.3	11.9	10.3	10.0
HS graduate	38.3	38.6	37.7	36.9	33.3	32.5
Some college	19.2	20.9	22.9	25.2	28.1	28.1

 Table 2

 Demographic Characteristics of Persons Aged 18 and Over by Business Cycle Period.

College graduate	12.1	13.9	16.9	17.5	22.2	23.4

Notes: Authors' calculations based on March Current Population Surveys. See text for further sample restrictions. Hispanics may be of any race. For consistency over time, age is top-coded at 80 years.

	Share of Population 18 and over		Percent w weeks la	Percent working 52 weeks last year		Average number of weeks worked last year		Percent working 0 weeks last year	
Age Group	1979	2001	1979	2001	1979	2001	1979	2001	
18 to 24	18.54	13.69	38.15	42.99	34.28	32.19	9.32	19.52	
25 to 29	11.99	8.92	66.17	74.12	45.03	44.83	3.94	7.33	
30 to 34	10.80	9.75	73.96	80.34	47.28	46.63	3.16	6.02	
35 to 39	8.78	11.03	76.35	81.11	47.84	47.03	3.12	5.47	
40 to 44	7.54	11.49	76.25	79.85	46.98	46.14	4.71	7.51	
45 to 49	7.54	10.21	76.04	78.36	46.23	45.39	5.90	8.86	
50 to 54	7.83	8.80	73.40	76.11	44.84	44.00	8.54	11.68	
55 to 59	7.47	6.58	67.27	67.66	40.97	39.42	15.67	19.43	
60 to 64	6.16	4.97	50.13	49.54	32.58	29.87	29.17	35.60	
65 to 69	5.28	4.57	20.65	24.28	15.57	16.36	61.06	61.57	
70 to 74	3.65	3.84	11.13	14.28	9.36	9.84	74.71	76.44	
75 to 79	2.38	3.08	7.52	9.42	6.38	6.33	82.96	85.27	
80 and older Total (18 and	2.03	3.06	3.69	5.17	2.92	3.41	92.65	92.22	
over)	100.00	100.00	56.80	62.18	38.04	37.46	17.06	21.56	

Table 3aWeeks Working by Age Group for Men, 1978 and 2000

Notes: Authors' calculations based on March Current Population Surveys for 1979 and 2001. We exclude men in the military and men who live in group quarters.

	Share of Population 18 and over		Percent working 52 weeks last year		Average number of weeks worked last year		Percent working 0 weeks last year	
Age Group	1979	2001	1979	2001	1979	2001	1979	2001
18 to 24	17.71	12.73	28.72	39.92	27.80	30.60	21.78	23.16
25 to 29	11.33	8.49	37.31	56.91	29.79	36.81	26.73	19.36
30 to 34	10.18	9.45	34.72	58.39	27.37	37.40	32.58	18.77
35 to 39	8.48	10.47	37.98	59.97	29.00	37.48	30.95	20.04
40 to 44	7.25	10.88	40.60	62.41	29.42	38.43	32.16	19.09
45 to 49	7.11	9.82	42.05	63.73	28.88	38.95	35.06	18.72
50 to 54	7.53	8.57	39.28	60.58	27.02	37.29	39.09	22.47
55 to 59	7.28	6.55	36.32	50.86	24.16	31.20	46.66	34.29
60 to 64	6.22	5.30	25.31	33.90	17.83	21.79	58.93	52.96
65 to 69	5.86	4.66	9.48	13.91	7.99	10.05	79.20	76.11
70 to 74	4.53	4.50	4.52	7.16	4.01	5.63	89.01	84.77
75 to 79	3.23	4.05	2.54	3.92	2.40	2.65	93.78	93.20
80 and older	3.30	4.53	0.86	1.26	0.84	1.01	97.64	97.36
Total (18 and								
over)	100.00	100.00	30.05	46.26	23.39	29.75	42.58	34.91

Table 3bWeeks Working by Age Group for Women, 1978 and 2000

Notes: Authors' calculations based on March Current Population Surveys for 1979 and 2001. We exclude women in the military and women who live in group quarters.

			Obse	rved Nonemp	loyment Time P	eriod	
		1	2	3	4	5	6
		Peak 78-	Trough 82-	Peak 88-	Trough 91-	Peak 99-	Trough 01-
	Г	79	83	89	92	00	02
	Peak 78-79	4.17	6.45	5.86	7.2	6.95	8.03
	Trough 82- 83	4.07	6.39	5.78	7.14	6.85	7.94
Demographic Weights	Peak 88-89	4.08	6.37	5.78	7.14	6.86	7.96
vveights	Trough 91- 92	4.12	6.37	5.8	7.16	6.92	8.01
	Peak 99-00	4.28	6.4	5.92	7.25	7.11	8.17
	Trough 01- 02	4.3	6.42	5.94	7.27	7.14	8.19

Table 4a: Percent of Men Aged 30 to 50 with 52 Weeks of Nonemployment: Observed Labor Force Status by TimePeriod Reweighted by Age

Source: Based on authors' calculations using March CPS data. Estimates used to create the weights include a fourthorder polynomial in age.

			Observed No	nemployment	Time Period		
		1 Peak 78- 79	2 Trough 82- 83	3 Peak 88- 89	4 Trough 91- 92	5 Peak 99- 00	6 Trough 01- 02
	Peak 78-79	31.46	28.7	22.12	21.75	19.31	21.35
	l rough 82- 83	31.36	28.57	22.03	21.72	19.35	21.4
Demographic Weights	Peak 88-89 Trough 91-	31.31	28.52	21.96	21.63	19.34	21.35
	92 92	31.33	28.57	21.96	21.6	19.32	21.31
	Peak 99-00 Trough 01-	31.46	28.78	21.98	21.5	19.26	21.14
	02	31.53	28.86	22.04	21.54	19.25	21.12

 Table 4b: Percent of Women Aged 30 to 50 with 52 Weeks of Nonemployment: Observed Labor Force Status by Time

 Period Reweighted by Age

Source: Based on authors' calculations using March CPS data. Estimates used to create the weights include a fourthorder polynomial in age.

			Obse	rved Nonemp	loyment Time P	eriod	
		1	2	3	4	5	6
		Peak 78- 79	1 rough 82- 83	Peak 88- 89	1 rough 91- 92	Peak 99- 00	1 rough 01- 02
	Peak 78-79	4.17	6.34	5.63	6.86	6.57	7.6
	Trough 82- 83	4.12	6.39	5.64	6.91	6.53	7.59
Demographic Weights	Peak 88-89	4.19	6.5	5.78	7.05	6.63	7.69
roigne	1 rough 91- 92	4.26	6.55	5.85	7.16	6.7	7.77
	Peak 99-00	4.58	6.89	6.24	7.59	7.11	8.16
	-100gH 01- 02	4.63	6.95	6.34	7.69	7.14	8.19

Table 5a: Percent of Men Aged 30 to 50 with 52 Weeks of Nonemployment: Observed Labor Force Status by TimePeriod Reweighted by Age, Race, and Ethnicity

Source: Based on authors' calculations using March CPS data. Estimates used to create the weights include a fourthorder polynomial in age, indicators for race is African American and race is other, an indicator for Hispanic, and a full set of pair-wise interaction terms.

			Obse	rved Nonemp	loyment Time P	eriod	
		1 Peak 78-	2 Trough 82-	3 Peak 88-	4 Trough 91-	5 Peak 99-	6 Trough 01-
	Г	79	83	89	92	00	02
	Peak 78-79	31.46	28.54	21.75	21.19	18.46	20.17
	Trough 82- 83	31.46	28.57	21.87	21.41	18.67	20.42
Demographic Weights	Peak 88-89	31.52	28.68	21.96	21.52	18.79	20.51
J	92	31.6	28.81	22.05	21.6	18.85	20.55
	Peak 99-00	32.05	29.52	22.55	22.11	19.26	20.92
	1 rough 01- 02	32.32	29.81	22.84	22.42	19.46	21.12

Table 5b: Percent of Women Aged 30 to 50 with 52 Weeks of Nonemployment: Observed Labor Force Status byTime Period Reweighted by Age, Race, and Ethnicity

Source: Based on authors' calculations using March CPS data. Estimates used to create the weights include a fourth-order polynomial in age, indicators for race is African American and race is other, an indicator for Hispanic, and a full set of pair-wise interaction terms.

			Obse	rved Nonemp	loyment Time P	eriod	
	-	1 Peak 78- 79	2 Trough 82- 83	3 Peak 88- 89	4 Trough 91- 92	5 Peak 99- 00	6 Trough 01- 02
	Peak 78-79	4.17	6.1	5.03	5.92	5.41	6.05
	1 rough 82- 83	4.34	6.39	5.2	6.15	5.53	6.28
Demographic Weights	Peak 88-89 Trough 91-	4.89	7.12	5.78	6.79	6.06	6.88
	92	5.22	7.48	6.08	7.16	6.35	7.77
	Peak 99-00 Trough 01-	5.99	8.31	6.87	8.03	7.11	7.21
	02	6.11	8.49	7.1	8.25	7.14	8.19

Table 6a: Percent of Men Aged 30 to 50 with 52 Weeks of Nonemployment: Observed Labor Force Status by Time Period Reweighted by Age, Race, Ethnicity, and Marital Status

Source: Based on authors' calculations using March CPS data. Estimates used to create weights include a fourthorder polynomial in age, indicators for race is African-American and race is other, an indicator for Hispanic, marital status indicators for married spouse absent, widowed, divorced, separated, and never married, and a full set of pairwise interaction terms.

			Obse	rved Nonemp	loyment Time P	eriod	
		1 Peak 78- 79	2 Trough 82- 83	3 Peak 88- 89	4 Trough 91- 92	5 Peak 99- 00	6 Trough 01- 02
	Peak 78-79	31.46	29.05	22.45	21.79	19.31	21.02
	1 rough 82- 83	30.82	28.57	22.21	21.69	19.17	20.92
Demographic Weights	Peak 88-89 Trough 91-	30.35	28.22	21.96	21.56	19.02	20.77
	92 92	30.25	28.21	21.95	21.6	18.95	20.69
	Peak 99-00 Trough 01-	30.46	28.74	22.31	21.99	19.26	21
	02	30.61	28.95	22.53	22.22	19.39	21.12

Table 6b: Percent of Women Aged 30 to 50 with 52 Weeks of Nonemployment: Observed Labor Force Status by Time Period Reweighted by Age, Race, Ethnicity, and Marital Status

Source: Based on authors' calculations using March CPS data. Estimates used to create weights include a fourthorder polynomial in age, indicators for race is African-American and race is other, an indicator for Hispanic, marital status indicators for married spouse absent, widowed, divorced, separated, and never married, and a full set of pairwise interaction terms.

	Observed Nonemployment Time Period							
	1 Peak 78- 79	2 Trough 82- 83	3 Peak 88- 89	4 Trough 91- 92	5 Peak 99- 00	6 Trough 01- 02		
Peak 78-79 Trough 82-	4.17	6.54	6.1	7.41	7.67	8.16		
Peak 88-89	4.00	6.61	5.78	6.93	6.85	7.58		
1 rough 91- 92	4.43	6.81	5.94	7.16	6.94	7.72		
Peak 99-00 Trough 01-	4.66	6.89	6.11	7.47	7.11	7.98		
	Peak 78-79 Trough 82- 83 Peak 88-89 Trough 91- 92 Peak 99-00 Trough 01- 02	1 Peak 78-79 Peak 78-79 4.17 Trough 82- 4.08 Peak 88-89 4.24 Trough 91- 92 Peak 99-00 4.66 Trough 01- 02 4.86	1 2 Peak 78-79 Trough 82-79 79 83 Peak 78-79 4.17 Trough 82-83 4.08 6.54 6.39 Peak 88-89 4.24 Trough 91-92 4.43 Peak 99-00 4.66 Trough 01-02 4.86	1 2 3 Peak 78-79 1700gh 82- 83 Peak 88- 89 Peak 78-79 4.17 6.54 6.1 Trough 82- 83 4.08 6.39 5.75 Peak 88-89 4.24 6.61 5.78 Trough 91- 92 4.43 6.81 5.94 Peak 99-00 4.66 6.89 6.11 Trough 01- 02 4.86 7.16 6.39	1 2 3 4 Peak 78-79 Trough 82- 79 Peak 88- 83 Peak 88- 89 Trough 91- 92 Peak 78-79 4.17 6.54 6.1 7.41 Trough 82- 83 4.08 6.39 5.75 6.96 Peak 88-89 4.24 6.61 5.78 6.93 Trough 91- 92 4.43 6.81 5.94 7.16 Peak 99-00 4.66 6.89 6.11 7.47 Trough 01- 02 4.86 7.16 6.39 7.71	1 2 3 4 5 Peak 78-79 Trough 82- 79 Peak 88- 83 Peak 88- 99 Trough 91- 92 Peak 99- 00 Peak 78-79 4.17 6.54 6.1 7.41 7.67 Trough 82- 83 4.08 6.39 5.75 6.96 7.1 Peak 88-89 4.24 6.61 5.78 6.93 6.85 Trough 91- 92 4.43 6.81 5.94 7.16 6.94 Peak 99-00 4.66 6.89 6.11 7.47 7.11 Trough 01- 02 4.86 7.16 6.39 7.71 7.32		

Table 7a: Percent of Men Aged 30 to 50 with 52 Weeks of Nonemployment: Observed Labor Force Status by Time Period Reweighted by Age, Race, Ethnicity, Marital Status, and Education

Source: Based on authors' calculations using March CPS data. Estimates used to create weights include a fourthorder polynomial in age, indicators for race is African-American and race is other, an indicator for Hispanic, marital status indicators for married spouse absent, widowed, divorced, separated, and never married, indicators for education level elementary, some high school, some college, and college graduate, and a full set of pair-wise interaction terms.

		Observed Nonemployment Time Period					
		1 Peak 78- 79	2 Trough 82- 83	3 Peak 88- 89	4 Trough 91- 92	5 Peak 99- 00	6 Trough 01- 02
	Peak 78-79	31.46	30.44	24.99	25.45	23.06	25.37
Demographic Weights	83 83	29.73	28.57	23.45	23.78	21.63	23.9
	Peak 88-89 Trough 91-	28.1	26.85	21.96	22.15	20.37	22.41
	92	27.55	26.3	21.45	21.6	19.82	21.82
	Peak 99-00 Trough 01-	26.49	25.44	20.69	20.72	19.26	21.21
	02	26.28	25.26	20.56	20.59	19.16	21.12

Table 7b: Percent of Women Aged 30 to 50 with 52 Weeks of Nonemployment: Observed Labor Force Status by Time Period Reweighted by Age, Race, Ethnicity, Marital Status, and Education

Source: Based on authors' calculations using March CPS data. Estimates used to create weights include a fourthorder polynomial in age, indicators for race is African-American and race is other, an indicator for Hispanic, marital status indicators for married spouse absent, widowed, divorced, separated, and never married, indicators for education level elementary, some high school, some college, and college graduate, and a full set of pair-wise interaction terms.

Period of Change	Observed Change in Nonemployment	Period of Observed Nonemployment	Demographic Characteristics Used to Construct Weights:	Change in Nonemployment Attributed to Demographic Changes	% of Change in Nonemployment Attributed to Demographic Changes
		1978-79		0.11	3.7
		1999-2000	age	0.16	5.4
		1978-79	age, race, and ethnicity	0.41	13.9
		1999-2000		0.54	18.4
1978-79 to 1999-2000	2.94	1978-79	age, race,	1.82	61.9
		1999-2000	ethnicity, and marital status	1.70	57.8
		1978-79	age, race, ethnicity,	0.49	16.7
		1999-2000	marital status, and education	-0.56	-19.0
		1982-83		0.03	1.7
	1.8	2001-02	age	0.25	13.9
		1982-83	age, race, and ethnicity	0.56	31.1
1000 00 1-		2001-02		0.60	33.3
1982-83 to 2001-02		1982-83	age, race, ethnicity, and marital status	2.10	116.7
		2001-02		1.91	106.1
		1982-83	age, race, ethnicity,	0.77	42.8
		2001-02	marital status, and education	0.49	27.2

Table 8a: Percent of Men Aged 30 to 50 with 52 Weeks of Nonemployment, Observed Change and Change Predicted by Demographic Characteristics for Select Time Periods

Source: Authors' calculations based on the estimates presented in Tables 4a, 5a, 6a, and 7a.

Period of Change	Observed Change in Nonemployment	Period of Observed Nonemployment	Demographic Characteristics Used to Construct Weights:	Change in Nonemployment Attributed to Demographic Changes	% of Change in Nonemployment Attributed to Demographic Changes
		1978-79	age	0.00	0.0
		1999-2000		-0.05	0.4
		1978-79	age, race, and	0.59	-4.8
1079 70 to		1999-2000	ethnicity	0.80	-6.6
1999-2000	-12.2	1978-79	age, race, ethnicity, and marital status	-1.00	8.2
		1999-2000		-0.05	0.4
		1978-79	age, race, ethnicity,	-4.97	40.7
		1999-2000	marital status, and education	-3.80	31.1
		1982-83	age	0.29	-3.9
	-7.45	2001-02		-0.28	3.8
		1982-83	age, race, and	1.24	-16.6
4000 00 1-		2001-02	ethnicity	0.70	-9.4
1982-83 to 2001-02		1982-83	age, race, ethnicity and	0.38	-5.1
		2001-02	marital status	0.20	-2.7
		1982-83	age, race, ethnicity,	-3.31	44.4
		2001-02	marital status, and education	-2.78	37.3

Table 8b: Percent of Women Aged 30 to 50 with 52 Weeks of Nonemployment, Observed Change and
Change Predicted by Demographic Characteristics for Select Time Periods

Source: Authors' calculations based on the estimates presented in Tables 4b, 5b, 6b, and 7b.

Working Paper Series

A series of research studies on regional economic issues relating to the Seventh Federal Reserve District, and on financial and economic topics.

Does Bank Concentration Lead to Concentration in Industrial Sectors? Nicola Cetorelli	WP-01-01
On the Fiscal Implications of Twin Crises Craig Burnside, Martin Eichenbaum and Sergio Rebelo	WP-01-02
Sub-Debt Yield Spreads as Bank Risk Measures Douglas D. Evanoff and Larry D. Wall	WP-01-03
Productivity Growth in the 1990s: Technology, Utilization, or Adjustment? Susanto Basu, John G. Fernald and Matthew D. Shapiro	WP-01-04
Do Regulators Search for the Quiet Life? The Relationship Between Regulators and The Regulated in Banking <i>Richard J. Rosen</i>	WP-01-05
Learning-by-Doing, Scale Efficiencies, and Financial Performance at Internet-Only Banks <i>Robert DeYoung</i>	WP-01-06
The Role of Real Wages, Productivity, and Fiscal Policy in Germany's Great Depression 1928-37 Jonas D. M. Fisher and Andreas Hornstein	WP-01-07
Nominal Rigidities and the Dynamic Effects of a Shock to Monetary Policy <i>Lawrence J. Christiano, Martin Eichenbaum and Charles L. Evans</i>	WP-01-08
Outsourcing Business Service and the Scope of Local Markets Yukako Ono	WP-01-09
The Effect of Market Size Structure on Competition: The Case of Small Business Lending Allen N. Berger, Richard J. Rosen and Gregory F. Udell	WP-01-10
Deregulation, the Internet, and the Competitive Viability of Large Banks and Community Banks <i>Robert DeYoung and William C. Hunter</i>	WP-01-11
Price Ceilings as Focal Points for Tacit Collusion: Evidence from Credit Cards Christopher R. Knittel and Victor Stango	WP-01-12
Gaps and Triangles Bernardino Adão, Isabel Correia and Pedro Teles	WP-01-13
A Real Explanation for Heterogeneous Investment Dynamics Jonas D.M. Fisher	WP-01-14
Recovering Risk Aversion from Options Robert R. Bliss and Nikolaos Panigirtzoglou	WP-01-15
Economic Determinants of the Nominal Treasury Yield Curve Charles L. Evans and David Marshall	WP-01-16

Price Level Uniformity in a Random Matching Model with Perfectly Patient Traders Edward J. Green and Ruilin Zhou	WP-01-17
Earnings Mobility in the US: A New Look at Intergenerational Inequality <i>Bhashkar Mazumder</i>	WP-01-18
The Effects of Health Insurance and Self-Insurance on Retirement Behavior <i>Eric French and John Bailey Jones</i>	WP-01-19
The Effect of Part-Time Work on Wages: Evidence from the Social Security Rules Daniel Aaronson and Eric French	WP-01-20
Antidumping Policy Under Imperfect Competition Meredith A. Crowley	WP-01-21
Is the United States an Optimum Currency Area? An Empirical Analysis of Regional Business Cycles <i>Michael A. Kouparitsas</i>	WP-01-22
A Note on the Estimation of Linear Regression Models with Heteroskedastic Measurement Errors Daniel G. Sullivan	WP-01-23
The Mis-Measurement of Permanent Earnings: New Evidence from Social Security Earnings Data <i>Bhashkar Mazumder</i>	WP-01-24
Pricing IPOs of Mutual Thrift Conversions: The Joint Effect of Regulation and Market Discipline <i>Elijah Brewer III, Douglas D. Evanoff and Jacky So</i>	WP-01-25
Opportunity Cost and Prudentiality: An Analysis of Collateral Decisions in Bilateral and Multilateral Settings Herbert L. Baer, Virginia G. France and James T. Moser	WP-01-26
Outsourcing Business Services and the Role of Central Administrative Offices <i>Yukako Ono</i>	WP-02-01
Strategic Responses to Regulatory Threat in the Credit Card Market* Victor Stango	WP-02-02
The Optimal Mix of Taxes on Money, Consumption and Income <i>Fiorella De Fiore and Pedro Teles</i>	WP-02-03
Expectation Traps and Monetary Policy Stefania Albanesi, V. V. Chari and Lawrence J. Christiano	WP-02-04
Monetary Policy in a Financial Crisis Lawrence J. Christiano, Christopher Gust and Jorge Roldos	WP-02-05
Regulatory Incentives and Consolidation: The Case of Commercial Bank Mergers and the Community Reinvestment Act <i>Raphael Bostic, Hamid Mehran, Anna Paulson and Marc Saidenberg</i>	WP-02-06

Technological Progress and the Geographic Expansion of the Banking Industry Allen N. Berger and Robert DeYoung	WP-02-07
Choosing the Right Parents: Changes in the Intergenerational Transmission of Inequality — Between 1980 and the Early 1990s <i>David I. Levine and Bhashkar Mazumder</i>	WP-02-08
The Immediacy Implications of Exchange Organization James T. Moser	WP-02-09
Maternal Employment and Overweight Children Patricia M. Anderson, Kristin F. Butcher and Phillip B. Levine	WP-02-10
The Costs and Benefits of Moral Suasion: Evidence from the Rescue of Long-Term Capital Management <i>Craig Furfine</i>	WP-02-11
On the Cyclical Behavior of Employment, Unemployment and Labor Force Participation <i>Marcelo Veracierto</i>	WP-02-12
Do Safeguard Tariffs and Antidumping Duties Open or Close Technology Gaps? Meredith A. Crowley	WP-02-13
Technology Shocks Matter Jonas D. M. Fisher	WP-02-14
Money as a Mechanism in a Bewley Economy Edward J. Green and Ruilin Zhou	WP-02-15
Optimal Fiscal and Monetary Policy: Equivalence Results Isabel Correia, Juan Pablo Nicolini and Pedro Teles	WP-02-16
Real Exchange Rate Fluctuations and the Dynamics of Retail Trade Industries on the U.SCanada Border <i>Jeffrey R. Campbell and Beverly Lapham</i>	WP-02-17
Bank Procyclicality, Credit Crunches, and Asymmetric Monetary Policy Effects: A Unifying Model Robert R. Bliss and George G. Kaufman	WP-02-18
Location of Headquarter Growth During the 90s Thomas H. Klier	WP-02-19
The Value of Banking Relationships During a Financial Crisis: Evidence from Failures of Japanese Banks Elijah Brewer III, Hesna Genay, William Curt Hunter and George G. Kaufman	WP-02-20
On the Distribution and Dynamics of Health Costs Eric French and John Bailey Jones	WP-02-21
The Effects of Progressive Taxation on Labor Supply when Hours and Wages are Jointly Determined Daniel Aaronson and Eric French	WP-02-22

Inter-industry Contagion and the Competitive Effects of Financial Distress Announcements: Evidence from Commercial Banks and Life Insurance Companies <i>Elijah Brewer III and William E. Jackson III</i>	WP-02-23
State-Contingent Bank Regulation With Unobserved Action and Unobserved Characteristics David A. Marshall and Edward Simpson Prescott	WP-02-24
Local Market Consolidation and Bank Productive Efficiency Douglas D. Evanoff and Evren Örs	WP-02-25
Life-Cycle Dynamics in Industrial Sectors. The Role of Banking Market Structure <i>Nicola Cetorelli</i>	WP-02-26
Private School Location and Neighborhood Characteristics Lisa Barrow	WP-02-27
Teachers and Student Achievement in the Chicago Public High Schools Daniel Aaronson, Lisa Barrow and William Sander	WP-02-28
The Crime of 1873: Back to the Scene <i>François R. Velde</i>	WP-02-29
Trade Structure, Industrial Structure, and International Business Cycles Marianne Baxter and Michael A. Kouparitsas	WP-02-30
Estimating the Returns to Community College Schooling for Displaced Workers Louis Jacobson, Robert LaLonde and Daniel G. Sullivan	WP-02-31
A Proposal for Efficiently Resolving Out-of-the-Money Swap Positions at Large Insolvent Banks <i>George G. Kaufman</i>	WP-03-01
Depositor Liquidity and Loss-Sharing in Bank Failure Resolutions George G. Kaufman	WP-03-02
Subordinated Debt and Prompt Corrective Regulatory Action Douglas D. Evanoff and Larry D. Wall	WP-03-03
When is Inter-Transaction Time Informative? Craig Furfine	WP-03-04
Tenure Choice with Location Selection: The Case of Hispanic Neighborhoods in Chicago Maude Toussaint-Comeau and Sherrie L.W. Rhine	WP-03-05
Distinguishing Limited Commitment from Moral Hazard in Models of Growth with Inequality* Anna L. Paulson and Robert Townsend	WP-03-06
Resolving Large Complex Financial Organizations Robert R. Bliss	WP-03-07

The Case of the Missing Productivity Growth: Or, Does information technology explain why productivity accelerated in the United States but not the United Kingdom? <i>Susanto Basu, John G. Fernald, Nicholas Oulton and Sylaja Srinivasan</i>	WP-03-08
Inside-Outside Money Competition Ramon Marimon, Juan Pablo Nicolini and Pedro Teles	WP-03-09
The Importance of Check-Cashing Businesses to the Unbanked: Racial/Ethnic Differences <i>William H. Greene, Sherrie L.W. Rhine and Maude Toussaint-Comeau</i>	WP-03-10
A Structural Empirical Model of Firm Growth, Learning, and Survival <i>Jaap H. Abbring and Jeffrey R. Campbell</i>	WP-03-11
Market Size Matters Jeffrey R. Campbell and Hugo A. Hopenhayn	WP-03-12
The Cost of Business Cycles under Endogenous Growth Gadi Barlevy	WP-03-13
The Past, Present, and Probable Future for Community Banks Robert DeYoung, William C. Hunter and Gregory F. Udell	WP-03-14
Measuring Productivity Growth in Asia: Do Market Imperfections Matter? John Fernald and Brent Neiman	WP-03-15
Revised Estimates of Intergenerational Income Mobility in the United States Bhashkar Mazumder	WP-03-16
Product Market Evidence on the Employment Effects of the Minimum Wage Daniel Aaronson and Eric French	WP-03-17
Estimating Models of On-the-Job Search using Record Statistics Gadi Barlevy	WP-03-18
Banking Market Conditions and Deposit Interest Rates Richard J. Rosen	WP-03-19
Creating a National State Rainy Day Fund: A Modest Proposal to Improve Future State Fiscal Performance <i>Richard Mattoon</i>	WP-03-20
Managerial Incentive and Financial Contagion Sujit Chakravorti, Anna Llyina and Subir Lall	WP-03-21
Women and the Phillips Curve: Do Women's and Men's Labor Market Outcomes Differentially Affect Real Wage Growth and Inflation? <i>Katharine Anderson, Lisa Barrow and Kristin F. Butcher</i>	WP-03-22
Evaluating the Calvo Model of Sticky Prices Martin Eichenbaum and Jonas D.M. Fisher	WP-03-23

The Growing Importance of Family and Community: An Analysis of Changes in the Sibling Correlation in Earnings <i>Bhashkar Mazumder and David I. Levine</i>	WP-03-24
Should We Teach Old Dogs New Tricks? The Impact of Community College Retraining on Older Displaced Workers <i>Louis Jacobson, Robert J. LaLonde and Daniel Sullivan</i>	WP-03-25
Trade Deflection and Trade Depression Chad P. Brown and Meredith A. Crowley	WP-03-26
China and Emerging Asia: Comrades or Competitors? Alan G. Ahearne, John G. Fernald, Prakash Loungani and John W. Schindler	WP-03-27
International Business Cycles Under Fixed and Flexible Exchange Rate Regimes <i>Michael A. Kouparitsas</i>	WP-03-28
Firing Costs and Business Cycle Fluctuations Marcelo Veracierto	WP-03-29
Spatial Organization of Firms Yukako Ono	WP-03-30
Government Equity and Money: John Law's System in 1720 France François R. Velde	WP-03-31
Deregulation and the Relationship Between Bank CEO Compensation and Risk-Taking Elijah Brewer III, William Curt Hunter and William E. Jackson III	WP-03-32
Compatibility and Pricing with Indirect Network Effects: Evidence from ATMs Christopher R. Knittel and Victor Stango	WP-03-33
Self-Employment as an Alternative to Unemployment <i>Ellen R. Rissman</i>	WP-03-34
Where the Headquarters are – Evidence from Large Public Companies 1990-2000 <i>Tyler Diacon and Thomas H. Klier</i>	WP-03-35
Standing Facilities and Interbank Borrowing: Evidence from the Federal Reserve's New Discount Window Craig Furfine	WP-04-01
Netting, Financial Contracts, and Banks: The Economic Implications William J. Bergman, Robert R. Bliss, Christian A. Johnson and George G. Kaufman	WP-04-02
Real Effects of Bank Competition Nicola Cetorelli	WP-04-03
Finance as a Barrier To Entry: Bank Competition and Industry Structure in Local U.S. Markets? <i>Nicola Cetorelli and Philip E. Strahan</i>	WP-04-04

The Dynamics of Work and Debt Jeffrey R. Campbell and Zvi Hercowitz	WP-04-05
Fiscal Policy in the Aftermath of 9/11 Jonas Fisher and Martin Eichenbaum	WP-04-06
Merger Momentum and Investor Sentiment: The Stock Market Reaction To Merger Announcements <i>Richard J. Rosen</i>	WP-04-07
Earnings Inequality and the Business Cycle Gadi Barlevy and Daniel Tsiddon	WP-04-08
Platform Competition in Two-Sided Markets: The Case of Payment Networks <i>Sujit Chakravorti and Roberto Roson</i>	WP-04-09
Nominal Debt as a Burden on Monetary Policy Javier Díaz-Giménez, Giorgia Giovannetti, Ramon Marimon, and Pedro Teles	WP-04-10
On the Timing of Innovation in Stochastic Schumpeterian Growth Models Gadi Barlevy	WP-04-11
Policy Externalities: How US Antidumping Affects Japanese Exports to the EU Chad P. Bown and Meredith A. Crowley	WP-04-12
Sibling Similarities, Differences and Economic Inequality Bhashkar Mazumder	WP-04-13
Determinants of Business Cycle Comovement: A Robust Analysis Marianne Baxter and Michael A. Kouparitsas	WP-04-14
The Occupational Assimilation of Hispanics in the U.S.: Evidence from Panel Data <i>Maude Toussaint-Comeau</i>	WP-04-15
Reading, Writing, and Raisinets ¹ : Are School Finances Contributing to Children's Obesity? <i>Patricia M. Anderson and Kristin F. Butcher</i>	WP-04-16
Learning by Observing: Information Spillovers in the Execution and Valuation of Commercial Bank M&As <i>Gayle DeLong and Robert DeYoung</i>	WP-04-17
Prospects for Immigrant-Native Wealth Assimilation: Evidence from Financial Market Participation Una Okonkwo Osili and Anna Paulson	WP-04-18
Institutional Quality and Financial Market Development: Evidence from International Migrants in the U.S. <i>Una Okonkwo Osili and Anna Paulson</i>	WP-04-19
Are Technology Improvements Contractionary? Susanto Basu, John Fernald and Miles Kimball	WP-04-20

The Minimum Wage and Restaurant Prices Daniel Aaronson, Eric French and James MacDonald	WP-04-21
Betcha can't acquire just one: merger programs and compensation <i>Richard J. Rosen</i>	WP-04-22
Not Working: Demographic Changes, Policy Changes, and the Distribution of Weeks (Not) Worked <i>Lisa Barrow and Kristin F. Butcher</i>	WP-04-23